Author's response to reviews

Title: Seroprevalence of Immunoglobulin G antibodies against pertussis toxin among asymptomatic medical students in the west of Iran: a cross sectional study

Authors:

Seyyed Hamid Hashemi (mitranjbar@yahoo.com)
Mitra Ranjbar (mitranjbar@yahoo.com)
Mehrdad Hajilooi (mitranjbar@yahoo.com)
Mohammad-Ali Seif-Rabiei (mitranjbar@yahoo.com)
Mahnaz Bolandi (mitranjbar@yahoo.com)
Javad Moghimi (mitranjbar@yahoo.com)

Version: 4 Date: 5 May 2009

Author's response to reviews:

Dear Editor

Many thanks for your comments. We went through all the comments and revised the manuscript point by point. Authors affiliations and email address were added. Abstract was corrected and abbreviation list and typos and grammatical errors were also corrected.

Regards,

Ranjbar, MD